Title: Assessing secondary attack rates among household contacts at the beginning of the influenza A (H1N1) pandemic in Ontario, Canada, April-June 2009: A prospective, observational study.

Authors:

Rachel Savage (rachel.savage@oahpp.ca)
Michael Whelan (Michael.Whelan@ontario.ca)
Ian Johnson (ian.johnson@oahpp.ca)
Elizabeth Rea (erea@toronto.ca)
Marie LaFreniere (marie.lafreniere@peelregion.ca)
Laura C Rosella (laura.rosella@oahpp.ca)
Freda Lam (freda.lam@oahpp.ca)
Tina Badiani (Tina.Badiani@ontario.ca)
Anne-Luise Winter (Anne-Luise.Winter@oahpp.ca)
Deborah J Carr (dcarr@county.oxford.on.ca)
Crystal Frenette (crystal.frenette@peelregion.ca)
Maureen Horn (maureen.horn@peelregion.ca)
Kathleen Dooling (kathleen.dooling@peelregion.ca)
Monali Varia (monali.varia@peelregion.ca)
Anne-Marie Holt (aholt@hkpr.on.ca)
Vidy Sunil (vsunil@hkpr.on.ca)
Catherine Griff (catherine.griff@mlhu.on.ca)
Eleanor Paget (eleanor.paget@mlhu.on.ca)
Michael King (kingm@sdhu.com)
John Barbaro (John.Barbaro@smdhu.org)
Natasha S Crowcroft (natasha.crowcroft@oahpp.ca)

Version: 3 Date: 21 March 2011

Author's response to reviews: see over
RE: Manuscript ID 6073301064814955

Assessing secondary attack rates among household contacts at the beginning of the influenza A (H1N1) pandemic in Ontario, Canada, April-June 2009: A prospective, observational study.

To Whom It May Concern,

Thank you for sharing the suggested formatting changes related to the above-named manuscript revision. Please see below for changes made:

<table>
<thead>
<tr>
<th>Reviewer Comment</th>
<th>Revision</th>
<th>Status</th>
</tr>
</thead>
<tbody>
<tr>
<td>Authors' contributions: Please indicate that all authors read and approved the final manuscript.</td>
<td>Added to end of authors’ contributions section.</td>
<td>Complete.</td>
</tr>
<tr>
<td>Highlighting/tracking: Please remove all highlighting and tracking from the manuscript.</td>
<td>Removed tracked changes.</td>
<td>Complete.</td>
</tr>
<tr>
<td>Figures: It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image.</td>
<td>The figure guidelines have been read and adhered to.</td>
<td>Complete.</td>
</tr>
<tr>
<td>Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions.</td>
<td>All final corrections/revisions have been made.</td>
<td>Complete.</td>
</tr>
</tbody>
</table>
If there are any additional questions and comments, these may still be forwarded as before to Rachel Savage at rachel.savage@oahpp.ca or by phone at 647-260-7403.

Sincerely,

Rachel Savage, MSc
Ontario Agency for Health Protection and Promotion
480 University Ave., Suite 300
Toronto, ON, Canada M5G 1V2
Tel: (647) 260-7403
E-mail: rachel.savage@oahpp.ca